ETH zürich

Aberrant computational mechanisms of social learning and decision-making in schizophrenia and borderline personality disorder

Journal Article

Author(s):

Henco, Lara; Diaconescu, Andreea O.; Lahnakoski, Juha M.; Brandi, Marie-Luise; Hörmann, Sophia; Hennings, Johannes; Hasan, Alkomiet; Papazova, Irina; Strube, Wolfgang; Bolis, Dimitris; Schilbach, Leonhard; Mathys, Christoph

Publication date: 2020-09-30

Permanent link: https://doi.org/10.3929/ethz-b-000448064

Rights / license: Creative Commons Attribution 4.0 International

Originally published in: PLoS Computational Biology 16(9), <u>https://doi.org/10.1371/journal.pcbi.1008162</u>



GOPEN ACCESS

Citation: Henco L, Diaconescu AO, Lahnakoski JM, Brandi M-L, Hörmann S, Hennings J, et al. (2020) Aberrant computational mechanisms of social learning and decision-making in schizophrenia and borderline personality disorder. PLoS Comput Biol 16(9): e1008162. https://doi.org/10.1371/journal. pcbi.1008162

Editor: Philipp Sterzer, Charite University Hospital, GERMANY

Received: March 11, 2020

Accepted: July 19, 2020

Published: September 30, 2020

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pcbi.1008162

Copyright: © 2020 Henco et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Applicable German federal law does not allow public archiving or peerto-peer sharing of individual raw data in this case. RESEARCH ARTICLE

Aberrant computational mechanisms of social learning and decision-making in schizophrenia and borderline personality disorder

Lara Henco^{1,2*}, Andreea O. Diaconescu^{3,4,5}, Juha M. Lahnakoski^{1,6,7}, Marie-Luise Brandi¹, Sophia Hörmann¹, Johannes Hennings⁸, Alkomiet Hasan^{9,10}, Irina Papazova⁹, Wolfgang Strube⁹, Dimitris Bolis^{1,11}, Leonhard Schilbach^{1,2,11,12*}, Christoph Mathys^{4,13,14*}

 Independent Max Planck Research Group for Social Neuroscience, Max Planck Institute of Psychiatry, Munich, Germany, 2 Graduate School for Systemic Neurosciences, Munich, Germany, 3 Department of Psychiatry (UPK), University of Basel, Basel, Switzerland, 4 Translational Neuromodeling Unit (TNU), Institute for Biomedical Engineering, University of Zurich and ETH Zurich, Zurich, Switzerland, 5 Krembil Centre for Neuroinformatics, Centre for Addiction and Mental Health (CAMH), University of Toronto, Canada, 6 Institute of Neuroscience and Medicine, Brain & Behaviour (INM-7), Research Centre Jülich, Jülich, Germany, 7 Institute of Systems Neuroscience, Medical Faculty, Heinrich Heine University Düsseldorf, Düsseldorf, Germany, 8 Department of Dialectical Behavioral Therapy, kbo-Isar-Amper-Klinikum Munich-East, Munich/Haar, Germany, 9 Department of Psychiatry and Psychotherapy, University Hospital Munich, LMU Munich, Munich, Germany, 10 Department of Psychiatry, Psychotherapy and Psychosomatic, University of Augsburg, Medical Faculty, Augsburg, Germany, 11 International Max Planck Research School for Translational Psychiatry (IMPRS-TP), Munich, Germany, 12 Medical Faculty, LMU Munich, Munich, Germany, 13 Scuola Internazionale Superiore di Studi Avanzati (SISSA), Trieste, Italy, 14 Interacting Minds Centre, Aarhus University, Aarhus, Denmark

These authors contributed equally to this work.
 * lara_henco@psych.mpg.de

Abstract

Psychiatric disorders are ubiquitously characterized by debilitating social impairments. These difficulties are thought to emerge from aberrant social inference. In order to elucidate the underlying computational mechanisms, patients diagnosed with major depressive disorder (N = 29), schizophrenia (N = 31), and borderline personality disorder (N = 31) as well as healthy controls (N = 34) performed a probabilistic reward learning task in which participants could learn from social and non-social information. Patients with schizophrenia and borderline personality disorder performed more poorly on the task than healthy controls and patients with major depressive disorder. Broken down by domain, borderline personality disorder patients performed better in the social compared to the non-social domain. In contrast, controls and major depressive disorder patients showed the opposite pattern and schizophrenia patients showed no difference between domains. In effect, borderline personality disorder patients gave up a possible overall performance advantage by concentrating their learning in the social at the expense of the non-social domain. We used computational modeling to assess learning and decision-making parameters estimated for each participant from their behavior. This enabled additional insights into the underlying learning and decision-making mechanisms. Patients with borderline personality disorder showed slower

However, the processed data underlying the main results of the study, together with the computational modeling code and behavioral analyses are available here: https://osf.io/8kfph/.

Funding: NO - The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

learning from social and non-social information and an exaggerated sensitivity to changes in environmental volatility, both in the non-social and the social domain, but more so in the latter. Regarding decision-making the modeling revealed that compared to controls and major depression patients, patients with borderline personality disorder and schizophrenia showed a stronger reliance on social relative to non-social information when making choices.

Depressed patients did not differ significantly from controls in this respect. Overall, our results are consistent with the notion of a general interpersonal hypersensitivity in borderline personality disorder and schizophrenia based on a shared computational mechanism characterized by an over-reliance on beliefs about others in making decisions and by an exaggerated need to make sense of others during learning specifically in borderline personality disorder.

Author summary

People suffering from psychiatric disorders frequently experience difficulties in social interaction, such as an impaired ability to use social signals to build representations of others and use these to guide behavior. Computtional models of learning and decisionmaking enable the characterization of individual patterns in learning and decision-making mechanisms that may be disorder-specific or disorder-general. We employed this approach to investigate the behavior of healthy participants and patients diagnosed with depression, schizophrenia, and borderline personality disorder while they performed a probabilistic reward learning task which included a social component. Patients with schizophrenia and borderline personality disorder performed more poorly on the task than controls and depressed patients. In addition, patients with borderline personality disorder concentrated their learning efforts more on the social compared to the non-social information. Computational modeling additionally revealed that borderline personality disorder patients showed a reduced flexibility in the weighting of newly obtained social and non-social information when learning about their predictive value. Instead, we found exaggerated learning of the volatility of social and non-social information. Additionally, we found a pattern shared between patients with borderline personality disorder and schizophrenia who both showed an over-reliance on predictions about social information during decision-making. Our modeling therefore provides a computational account of the exaggerated need to make sense of and rely on one's interpretation of others' behavior, which is prominent in both disorders.

Introduction

Impairments in social cognition are frequently experienced by people suffering from a psychiatric disorder. For instance, patients with major depressive disorder (MDD) and schizophrenia (SCZ) show a reduction in (social) reward sensitivity and motivation to engage in social interactions [1–5]. Despite high levels of social anhedonia, patients with SCZ show a tendency to over-interpret the meaning of social signals [6]. Individuals with borderline personality disorder (BPD) suffer from rapidly changing beliefs about others that polarise between approach and rejection [7]. Together, these impairments are associated with aberrant inferences/beliefs about oneself and the social environment.

In computational terms, the emergence of aberrant inference can be ascribed to an impaired ability to adjust learning in response to environmental changes [8]. Bayesian learning

models allow for a parsimonious algorithmic description of changes in beliefs relevant for accurate inference: belief updates can be written as a surprise signal (prediction error) weighted by a learning rate [9]. The learning rate depends on the ratio between the precision of the sensory data and the precision of the prior belief [10,11]. Whereas healthy participants increase their learning rate more strongly in volatile compared to stable environments [12,13], patients with autism do so less owing to an over-estimation of environmental volatility [8]. Impairments in the estimation of environmental volatility have also been studied as a mechanism for psychosis and SCZ [14–17] as well as MDD [18]. One recent study found that, unlike healthy controls, participants with BPD did not show an increase in learning when social and reward contingencies became volatile [19]. The authors suggested that this might be due to higher expected baseline volatility in participants with BPD. However, the computational model employed in that study did not explicitly model beliefs about volatility.

Adopting previous suggestions of aberrant volatility learning in psychiatric disorders and its role in impaired probability learning, the current study employed Bayesian hierarchical modeling to investigate probabilistic social inference in a volatile context across three major psychiatric disorders, which have previously been associated with social impairments: MDD, SCZ and BPD. Here, the current study investigated whether volatility and probability learning is equally affected when inferring on the hidden states of non-social and social outcomes across the three different disorders. We further asked whether aberrant social learning and decision-making were associated with differences in social anhedonia.

To this end, we adopted a probabilistic reward learning task (introduced in [20]), in which participants could learn from two types of information: non-social and social information. In order to probe the spontaneous rather than explicitly instructed use of social information as in previous social learning studies [12,13,21,22], we did not explicitly tell participants to learn about the social information. We used the hierarchical Gaussian filter (HGF; [10,11]) to obtain a profile of each participant's particular way of updating beliefs when receiving social and non-social information while making decisions in a volatile context. The HGF is a generic hierarchical Bayesian inference model for volatile environments with parameters that reflect individual variations in cognitive style. We went beyond other recent computational psychiatry studies using the HGF (e.g., [8,23–27]) in that we used two parallel HGF hierarchies for social and non-social aspects of the environment (cf. [20,29]). Our modeling framework was specifically designed also to quantify the relative weight participants afforded their beliefs about the predictive value of social compared to non-social information in decision-making.

Materials and methods

Ethics statement

All participants were naïve to the purpose of the experiment and provided written informed consent to take part in the study after a written and verbal explanation of the study procedure. The study was in line with the Declaration of Helsinki and approval for the experimental protocol was granted by the local ethics committee of the Medical Faculty of the Ludwig-Maximilians University of Munich.

Participants

Patients were recruited for the present study after an independent and experienced clinician diagnosed them using ICD-10 criteria for 1) a depressive episode (F32), schizophrenia (F20.0) and emotionally unstable personality disorder (F60.3). HC and patients with MDD were recruited through the Max Planck Institute of Psychiatry. Patients with SCZ were recruited at the Department of Psychiatry and Psychotherapy at the University Hospital Munich. Patients

with BPD were recruited at the kbo-Isar-Amper-Klinikum in Haar, Munich. Participants were chosen prior to analysis such that groups were matched for age ($\chi^2 = 5.302$, P = 0.151; Kruskal-Wallis one-way non-parametric ANOVA because of difference in age variance between groups, see S1 Table). Exclusion criteria were a history of neurological disease or injury, reported substance abuse at the time of the investigation, a history of electroconvulsive therapy, and diagnoses of comorbid personality disorder in the case of MDD and SCZ. Furthermore, 9 participants had to be excluded from the analysis due to one of the following reasons: unsaved data due to technical problems (1 HC, 2 BPD). Prior participation in another study which involved the same paradigm (1 HC), always picking the card with the higher reward value (1 HC), either following (1 SCZ) or going against (1 BPD) the gaze on more than 95% of trials (indicating a learning-free strategy), interruption of the task (1 SCZ), change to the diagnosis following study participation (1 MDD). The final sample consisted of 31 HC, 28 MDD, 29 SCZ and 28 BPD. We additionally acquired psychometric data (S1 Table) to further characterize the participants: All patients were asked to fill out questionnaires measuring autistic traits with the autism spectrum quotient (AQ [30]) and social anhedonia symptoms with the Anticipatory and Consummatory Interpersonal Pleasure Scale (ACIPS; [31]). We additionally assessed positive and negative symptoms using the Positive and Negative Syndrome Scale (PANSS [32]) and mood symptoms using the Calgary Depression Scale for Schizophrenia (CDSS [33]) in patients with SCZ. To assess the severity of Borderline Personality Disorder we used the short version of the Borderline Symptom List (BSL-23 [34]). Additional questionnaires were employed but analyzed within the scope of a different study and therefore not presented here. Demographic data as well as details regarding the medication can be seen in <u>S2 Table</u>.

Experimental paradigm and procedure

After giving informed consent, participants were seated in front of a computer screen in a quiet room where they received the task instructions. In the same probabilistic learning task as in [20,29], participants were asked to choose between one of two cards (blue or green) in order to maximize their score which was converted into a monetary reward $(1-6 \in)$ that was added to participants' compensation at the end of the task. An animated face was displayed between the cards, which first gazed down, then up towards the participant, before it shifted its gaze towards one of the cards (Fig 1A). The blue and green card appeared randomly on the left and right side from the face and participants responded using 'a' or 'l' on a German QWERTZ keyboard. When a response was logged within the allowed time (6000 ms), the chosen card was marked for 1000 ms until the outcome (correct: green check mark/wrong: red cross) was displayed for 1000 ms. When the correct card was chosen, the reward value (1-9) displayed on the card was added to the score. Participants were instructed that these values were not associated with the cards' winning probabilities, but that they might want to choose the card with the higher value if they were completely uncertain about the outcome. When the wrong card was chosen or participants failed to choose a card in the allotted time, the score remained unchanged. Participants were told that the cards had winning probabilities that changed in the course of the experiment but they were not informed about the systematic association between the face animation's gaze and the trial outcome. Specifically, they were not told that the probability with which the face animation pointed towards the winning card on a given trial varied systematically throughout the task according to the schedule given in Fig 1B. Instead, we simply told participants that the face was integrated into the task to make it more interesting. The probabilistic schedules for social and non-social information were independent from each other in order to estimate participant-specific learning rates separately for both types of information. In the first half of the experiment (trials 1-60), the card winning probabilities were





https://doi.org/10.1371/journal.pcbi.1008162.g001

stable, whereas in the second half (trials 61–120) they changed (volatile phase). The social cue had a stable contingency during trials 1–30 and trials 71–120, whereas contingency was volatile during trials 31–70. We used two types of schedules for the social cue which were each presented to half of the participants. In one schedule (depicted in Fig 1B), the probability of the social cue looking towards the winning card was 73% in the first stable phase (trial 1–30) and therefore started as congruent to the winning card (congruent-first). The second probability schedule was flipped, so that the probability of the social cue looking towards the winning card was 27% in the first stable phase (incongruent-first). In total, 15 control participants received the congruent-first schedule, 15 participants with MDD, 14 with SCZ and 15 with BPD. Positions of the cards on the screen (blue left or right) were determined randomly. The task was programmed and presented with PsyToolkit [35].

Computational modeling

Observing the observer. We modeled behavior in the 'observing the observer' (OTO) framework [36,37]. This entails a *response model*, which probabilistically predicts a

participant's choices based on their inferred beliefs, and a *perceptual model*, on which the response model depends because it describes the trajectories of participants' inferred beliefs based on experimental inputs. The OTO framework is conceptually very similar to the idea of *inverse reinforcement learning* [38].

Perceptual models. We used three different perceptual models in order to make inferences on the most likely mechanisms of learning in our paradigm. We used a Bayesian HGF as well as two non-Bayesian learning models, the Sutton K1 model [39] and a Rescorla Wagner model [40]. All three were implemented in the HGF toolbox, such that they perform parallel learning about the social (predictive value of gaze) and non-social (predictive value of card color) aspects of the task environment.

While the Rescorla Wagner learning model assumes fixed domain-specific learning rates for learning the social and non-social information, the Sutton K1 model assumes variable learning rates that are scaled by recent predicition errors. In contrast, the HGF takes into accout that beliefs have different degrees of uncertainty, scaling the learning rate dynamically as a function of uncertainty. In addition, the HGF assumes hierchical learning of different aspects of the environment. On the lowest level of the hierarchy, agents learn about concrete events (i.e. stimuli), whereas at higher levels of the hierarchy, agents learn about more abstract features of the environment, such as probabilistic associations between stimuli and how these change in time (i.e. volatility). Learning at every level is driven by a ratio between the precision of the input (from the level below) and the precision of prior beliefs.

The HGF is an inference model resulting from the inversion of a generative model in which states of the world are coupled in a three-level hierarchy: At the lowest level of the generative model, x_{1gaze} and x_{1card} represent the two inputs in a binary form (social cue: 1 = correct, 0 = incorrect; card outcome: 1 = blue wins, 0 = green wins). Level x_{2gaze} and x_{2card} represent the tendency of the gaze to be correct and the tendency of the blue card to win. State x_{2gaze} and x_{2card} evolve as first-order autoregressive (AR(1)) processes with a step size determined by the state at the third level. Level x_{3gaze} and x_{3card} represent the log-volatility of the two tendencies and also evolve as first-order autoregressive (AR(1)) processes. The probabilities of $x_{1gaze} = 1$ and $x_{1card} = 1$ are the logistic sigmoid transformations of x_{2gaze} and x_{2card} (Eq 1).

$$p(x_1^{(t)} = 1) = \frac{1}{1 + \exp(-x_2^{(t)})}$$
(1)

Participants' responses *y* were coded with respect to the congruency with the 'advice' (1 = follow; 0 = not follow) and were used to invert the model in order to infer the belief trajectories at all three levels *i* = 1,2,3.

On every trial k, the beliefs $\mu_i^{(k)}$ (and their precisions $\pi_i^{(k)}$) about the environmental states at the *i*-th level are updated via prediction errors $\delta_{i-1}^{(k)}$ from the level below weighted by a precision ratio $\psi_i^{(k)}$ (Eqs 2 and 3). This means that belief updates are larger (due to higher precision weights) when the precision of the posterior belief ($\pi_2^{(k)}$ or $\pi_3^{(k)}$) is low and the precision of the prediction $\hat{\pi}_2^{(k)}$ is high. Consequently, prediction errors are weighted more during phases of high volatility (cf. S1 Fig, panel C, dotted blue trajectory). For the analysis, we used $q(\psi_2^{(k)})$ (Eq 4), which is a transformation of $\psi_2^{(k)}$ (Eq 3) (cf. [41], supplementary material) that corrects for the sigmoid mapping between first and second level, effectively making $q(\psi_2^{(k)})$ an uncertainty (inverse precision) measure for first-level beliefs.

$$\Delta \mu_i^{(k)} \propto \psi_i^{(k)} \delta_{i-1}^{(k)} \ (i=2,3)$$
(2)

$$\psi_2^{(k)} = \frac{1}{\pi_2^{(k)}} \tag{3}$$

$$q(\psi_2^{(k)}) = \psi_2^{(k)} \, s(\mu_2^{(k)})(1 - s(\mu_2^{(k)})) \tag{4}$$

$$\psi_3^{(k)} = \frac{\hat{\pi}_2^{(k)}}{\pi_3^{(k)}} \tag{5}$$

Participant-specific parameters ω_{2card} and ω_{2gaze} represent the learning rates at the second level, i.e. the speed at which association strengths change. Correspondingly, ω_{3card} and ω_{3gaze} represent the learning rates of the volatilities.

Response models. In the response model a combined belief $b^{(t)}$ (Eq.6) was mapped onto decisions, which resulted from a combination of both the inferred prediction $\hat{\mu}_{1,gaze}^{(t)}$ that the face animation's gaze will go to the winning card and the inferred prediction $\hat{\mu}_{1,card}^{(t)}$ that the color of the card that the gaze went to would win (see example in S1 Fig). The inferred prediction $\hat{\mu}_{1,card}^{(t)}$ were weighted by $w_{gaze}^{(t)}$ and $w_{card}^{(t)}$ (Eqs.7 and 8), which are functions of the respective precisions ($\hat{\pi}_{1,gaze}^{(t)}$ and $\hat{\pi}_{1,card}^{(t)}$, Eqs.9 and 10). The precisions (Eqs.9 and 10) represent the inverse variances of a Bernoulli distribution of $\hat{\mu}_{1,gaze}^{(t)}$ and $\hat{\mu}_{1,card}^{(t)}$.

The constant parameter ζ represents the weight on the precision of the social prediction compared to the precision of the non-social prediction (Eq 7). In other words, this parameter describes the propensity to weight the social over the non-social information. We investigated the effect of varying the social weighting factor ζ , by simulating the combined belief $b^{(t)}$ (Eq 6) of agents with same perceptual parameters (fixed at prior values as depicted in S3 Table) but different ζ values.

$$b^{(t)} = w_{gaze}^{(t)} \,\hat{\mu}_{1,gaze}^{(t)} + w_{card}^{(t)} \,\hat{\mu}_{1,card}^{(t)} \tag{6}$$

$$w_{gaze}^{(t)} = \frac{\zeta \hat{\pi}_{1,gaze}^{(t)}}{\zeta \hat{\pi}_{1,gaze}^{(t)} + \hat{\pi}_{1,card}^{(t)}}$$
(7)

$$w_{card}^{(t)} = \frac{\hat{\pi}_{1,card}^{(t)}}{\zeta \hat{\pi}_{1,gaze}^{(t)} + \hat{\pi}_{1,card}^{(t)}}$$
(8)

$$\hat{\pi}_{1,gaze}^{(t)} = \frac{1}{\hat{\mu}_{1,gaze}^{(t)}(1 - \hat{\mu}_{1,gaze}^{(t)})}$$
(9)

$$\hat{\pi}_{1,card}^{(t)} = \frac{1}{\hat{\mu}_{1,card}^{(t)}(1 - \hat{\mu}_{1,card}^{(t)})}$$
(10)

In the response model, we used the combined belief $b^{(t)}$ (Eq 6) in a logistic sigmoid (softmax) function to model the probability $Prob_{gaze}^{(t)}$ (Eq 11). In this function, the belief was weighted by the predicted reward of the card when the advice is taken r_{gaze} or not $r_{notgaze}$ (Eq 11). We took account of possible subject-specific non-linear distortions in weighting the expected reward by using a weighted average (parameter η) of linear and logarithmic weighting of expected reward.

$$prob_{gaze} = p(y^{(t)} = 1) = \eta / (1 + \exp(-\gamma^{(t)}(r_{gaze}^{(t)}b^{(t)} - r_{notgaze}^{(t)}(1 - b^{(t)})))) + (1 - \eta) / (1 + \exp(-\gamma^{(t)}(\log(r_{gaze}^{(t)})b^{(t)} - \log(r_{notgaze}^{(t)})(1 - b^{(t)}))))$$

$$(11)$$

The mapping of beliefs onto actions varied as a function of the inverse decision temperature $\gamma^{(t)}$, where large $\gamma^{(t)}$ implied a high alignment between belief and choice (low decision noise) and a smaller $\gamma^{(t)}$ a low alignment between belief and choice (high decision noise). Our four different response models varied in terms of how $\gamma^{(t)}$ was defined. In response model 1, $\gamma^{(t)}$ was a combination of the log-volatility of the third level for both cues combined with constant participant-specific decision noise β (Eq 12). In response model 2, $\gamma^{(t)}$ was a combination of the log-volatility of the social cue and participant-specific decision noise (Eq 13) and in response model 3, $\gamma^{(t)}$ was a combination of the log-volatility of the social cue and participant-specific decision noise (Eq 13) and in response model 3, $\gamma^{(t)}$ was a combination of the log-volatility of the third level for the social noise (Eq 14). In model 4, $\gamma^{(t)}$ only included the participant-specific decision noise (Eq 15).

1)
$$\gamma^{(t)} = \beta \exp(-\hat{\mu}_{3,card}^{(t)} - \hat{\mu}_{3,gaze}^{(t)})$$
 (12)

2)
$$\gamma^{(t)} = \beta \exp(-\hat{\mu}_{3,gaze}^{(t)})$$
 (13)

3)
$$\gamma^{(t)} = \beta \exp(-\hat{\mu}_{3,card}^{(t)})$$
 (14)

$$4) \gamma^{(t)} = \beta \tag{15}$$

We used the HGF toolbox, version 4.1, which is part of the software package TAPAS (https://translationalneuromodeling.github.io/tapas) for parameter estimation. We fitted six alternate combinations of perceptual and response models, which were subjected to random-effects Bayesian Model Selection [42,43] (spm_BMS in SPM12; http://www.fil.ion.ucl.uk/spm). The HGF was combined with all four response models. The non-hierarchical models were combined with response model 4 only, owing to the lack of third-level belief trajectories. Details of the prior settings of all models can be seen in S3 Table.

We additionally included two non-learning models, which assume that participants repeat the actions that lead to reward and switch the strategy immediately after loss (Win-Stay-Lose-Shift) and a model that assumes random responding throughout the task (random-responding). These models were adapted from [44] and implemented in the HGF toolbox to calculate the log model evidence.

Model comparison and validity. The log model evidence (LME) for each participant and each model were subjected to Bayesian Model Selection [43,44] (spm_BMS in SPM12). This procedure estimates the expected posterior probabilities (EXP_P), i.e. the posterior probability of the prevalence of each model in the population, the exceedance probability (XP), i.e. the probability that a given model outperforms all others in the comparison, and the more conservative protected exceedance probability (PXP), which additionally considers the possibility that all models are equally good. We additionally performed within-participant model comparisons to identify the model with the highest LME for each participant (cf. <u>S4 Table</u>). This turned up 16 subjects where random responding had a higher model evidence than the overall winning model. Excluding these participants did not change the pattern of results (cf. <u>S7-S9</u> Tables).

Posterior predictive validity of model parameters and Parameter Recovery. To test the adequacy of the model, we simulated responses based on the estimated parameters from the winning model for each participant 10 times, resulting in 1160 simulations. As for the real data, we then calculated percentage of high reward probability choices for the two cues and the two phases, subjecting them to the same ANOVA that was performed with the real behavioral data. In this way, we checked whether the simulated responses produce the same group differences in response accuracy. In a second step, simulated responses were again used to invert the winning model to check whether model parameters could be recovered. For each subject, we calculated the average parameter values estimated from the simulated data and correlated (Pearson's correlations) them with the values estimated from the real data, which served as input for the simulation.

Regression-based choice sequence analysis. We ran a regression based choice sequence analysis in order to investigate adaptation to environmental volatility without needing to fit a learning model. According to the HGF, agents should increase their learning rate in more volatile environments, giving more weight to recent outcomes. In more stable environments, agents should adopt lower learning rates, affording less weight to recent outcomes in order to better filter out noise. To test this in a model-agnostic way, we implemented two general linear models (GLMs) with the responses *y* as outcome variables for the 'card' and 'gaze' space respectively (card: 1 = blue taken; 0 = green taken; gaze: 1 = follow; 0 = not follow). For each GLM, we included as predictors the outcomes of the past 5 trials (t-1:t-5: 1 = blue correct; 0 = green correct) and (t-1:t-5: 1 = gaze correct; 0 = gaze incorrect) as well as two predictors for the expected reward (1–9) for either card winning. We ran these GLMs for the stable and volatile phases separately. The slopes of the coefficient estimates of the past outcome predictors were taken as a model-agnostic readout of the 'learning rate', indiciating the degree to which more recent information is weighted. A t-test was applied to compare the difference between these slopes during stable and volatile phases of the task.

Statistical analysis. Performance (% correct responses) was subjected to a one-way ANOVA with group (HC vs. MDD vs. SCZ vs. BPD) and schedule (congruent first vs. incongruent first) as between-subject factors.

In order to understand task performance in a domain-specific way, we additionally calculated response accuracy based on the ground truth reward probability (high probability choices) of both cues (non-social and social) and both phases (stable and volatile). The proportion of high probability choices was subjected to a mixed ANOVA with Cue Type (Non-Social vs. Social) and Phase (Stable vs. Volatile) as within-subject factors and Group (HC vs. MDD vs. SCZ vs. BPD) and schedule (congruent first vs. incongruent first) as between-subject factors. Advice taking (advice followed or not on a given trial) was subjected to a mixed ANOVA with social accuracy (high vs. low) and schedule stability (stable vs. volatile) as within-subject factors. Group (HC vs. MDD vs. SCZ vs. BPD) and schedule (congruent first vs. incongruent first) were included as between-subject factors.

Mean precision weights on the second and third level ($q(\psi_2)$ and ψ_3) separately entered two mixed ANOVAs as dependent variables with schedule stability as a within-subject factor (stable vs.volatile), information type as within participants factor (social vs. non-social). The group (HC vs. MDD vs. SCZ vs. BPD) and schedule (congruent first vs. incongruent first) were between subject factors.

We subjected the posterior estimate for ζ to a one-way ANOVA with group (HC vs. MDD vs. SCZ vs. BPD) as between-subject factor and schedule (congruent first vs. incongruent first) as a covariate.

We hypothesized that social anhedonia (measured by the Anticipatory and Consummatory Interpersonal Pleasure Scale, ACIPS) would be associated with a reduction in learning in the social domain. To test this, we first performed a one-way ANOVA with ACIPS scores as dependent variable and group as the factor (HC vs. MDD vs. SCZ vs. BPD) followed by a multivariate regression with ACIPS as dependent variable and the social learning rates ω_{2gaze} and the weighting factor ζ as predictors of social learning and decision making. The group factor (HC vs. MDD vs. SCZ vs. BPD) was entered as covariate. This analysis was done for all participants who completed the ACIPS questionnaire (n = 106 of n_{total} = 116).

All ANOVA post hoc *t* tests were Bonferroni-corrected for multiple comparisons. All *p*-values are two-tailed with a significance threshold of p < .05. Statistical tests were performed using JASP (Version 0.9 2.0; https://jasp-stats.org/) or Matlab (Version 2018b; https://mathworks.com).

Results

Behavior

There was a significant difference between the groups in the overall performance, i.e. % of rewarded responses (F(3,108) = 7.504, p < 0.001, $\eta^2 = 0.167$): Post-hoc comparisons showed that both patients with SCZ and BPD performed significantly worse compared to HC and patients with MDD (SCZ–HC t = 3.781, $p_{\text{bonf}} = 0.002$, d = 0.994, SCZ–MDD t = 2.817, $p_{\text{bonf}} = 0.035$, d = 0.745, BPD–HC t = 3.732, $p_{\text{bonf}} = 0.002$, d = 0.979, BPD–MDD t = 2.78, $p_{\text{bonf}} = 0.038$, d = 0.732). There was no significant difference in performance between patients with BPD and SCZ (t = -0.01, $p_{\text{bonf}} = 1.000$, d = -0.003) nor between HC and patients with MDD (t = 0.88, $p_{\text{bonf}} = 1.000$, d = 0.240). Performance was not significantly affected by the schedule order (congruent first vs. incongruent first; F(1,108) = 0.027, p = 0.870, $\eta^2 = 0$) or its interaction with the patient groups (F(3,108) = 1.302, p = 0.278, $\eta^2 = 0.029$).

There was a significant difference between the groups in the overall proportion of trials where the better option was chosen, i.e. proportion of correct choices based on the ground truth reward probability of both cues (F(3,108) = 4.945, p = 0.003, $\eta^2 = 0.116$, Fig 2 and S6 Table). Post-hoc comparisons showed that both SCZ and BPD patients performed significantly worse compared to HC (SCZ–HC t = 3.373, $p_{bonf} = 0.006$, d = 0.313, BPD–HC t = 3.227, $p_{bonf} = 0.01$, d = 0.3) across domains (i.e., cue types) (Fig 2 and S6 Table). Overall, response accuracy did not significantly differ between cue types (F(1,108) = 2.577, p = 0.111, $\eta^2 = 0.019$), but



Fig 2. Proportion of trials where the better option was chosen. The better option was defined to be the one that according to the ground truth probability schedule was more likely to be rewarded. Importantly, 'better' choice according to color and according to social cue could be different in the same trial. Patients with SCZ and BPD showed poorer response quality in the task compared to HC. A Group x Cue interaction analysis showed that response quality was higher for the non-social cue for HC and MDD patients, whereas BPD patients showed the opposite pattern. Means are plotted with boxes marking 95% confidence intervals and vertical lines showing standard deviations. See also S6 Table.

https://doi.org/10.1371/journal.pcbi.1008162.g002



Fig 3. Behavioral results with regard to advice taking. All participants followed the advice significantly more during phases of high compared to low accuracy. The descriptive data shows a trend of BPD patients to follow the advice more during volatile phases of low accuracy (right most plot) but the interaction was not significant. Means are plotted with boxes marking 95% confidence intervals and vertical lines showing standard deviations.

https://doi.org/10.1371/journal.pcbi.1008162.g003

there was a significant Cue Type × Group interaction F(3,108) = 4.820, p = 0.003, $\eta^2 = 0.108$), with HC (Post-hoc: t(30) = -2.157, p_{bonf} = 0.039, d = -0.387) and MDD patients (Post-hoc: t(27) = -2.181, p_{bonf} = 0.038, d = -0.412) showing higher response accuracy with regard to the non-social cue, whereas BPD patients showed the opposite pattern (Post-hoc: t(27) = 2.058, p_{bonf} = 0.049, d = 0.389).

With regard to advice taking behavior during the different phases of the social schedule, we found a main effect of social accuracy (F(1,108) = 227.935, p < 0.001) whereby participants followed the gaze more during phases of high accuracy compared to phases of low accuracy (t = 14.94, $p_{bonf} < 0.001$) (Fig 3). Advice taking was not significantly affected by the schedule stability (F(1,108) = 0.503, p = 0.480), indicating that advice taking did not differ between stable and volatile phases. Advice taking was not significantly affected by an interaction between accuracy of the social information and Group (F(3,108) = 2.222, p = 0.09), or by an interaction between social accuracy, schedule stability and Group (F(3,108) = 1.47, p = 0.227).

Bayesian model comparison & validity

Model comparison showed that the HGF including subject specific decision noise as well as the volatility estimate $\hat{\mu}_{3,gaze}$ and $\hat{\mu}_{3,card}$ outperformed the other HGF models, the Rescorla Wagner and Sutton-K1 models with subject specific decision noise only as well as the WSLS and random responding models (PXP = 0.173; XP = 0.914). See Table 1 for further details and

Two to The Dury count of the second											
BMS	Model 1	Model 2	Model 3	Model 4	Model 5	Model 6	Model 7	Model 8			
EXP_R	0.304	0.181	0.073	0.019	0.066	0.210	0.072	0.076			
ХР	0.914	0.019	0	0	0	0.068	0	0			
РХР	0.173	0.119	0.117	0.117	0.117	0.122	0.117	0.117			

Tabl	e	1.	Bayesian	model	comparison	results.
------	---	----	----------	-------	------------	----------

Posterior model probabilities (EXP_R), Exceedance Probabilities (XP) and Protected Exceedance Probabilities (PXP). Model 1 refers to the HGF combined with response model 1, Model 2 refers to the HGF combined with response model 2, Model 3 refers to the HGF combined with response model 3, Model 4 refers to the HGF combined with response model 4, Model 5 refers to the Sutton K-1 Model combined with response model 4, Model 6 refers to the Rescorla Wagner Model combined with response model 4. Model 7 refers to the WSLS Model and Model 8 to the random Model.

https://doi.org/10.1371/journal.pcbi.1008162.t001

S4 Table for mean posterior parameter estimates. We repeated all modeling-based analyses with a reduced sample where all subjects were excluded for whom the random responding model outperformed the others (S4 Table). This did not lead to any qualitative change in results (S7–S9 Tables). We therefore report the results for the full sample here.

Posterior predictive validity of model parameters and parameter recovery

Subjecting the simulated behavioral readout of % of high probability choices to the same ANOVA as performed with the real behavioral data, we found that the model was able to reproduce the group differences that were observed in the real behavioral data (cf. S6 Table for comparison). The ANOVA of the simulated data showed that for response accuracy, as in the real data, there was a main effect of Group (F(3,108) = 6.755, p = 0.001, $\eta^2 = 0.149$) with posthoc t tests revealing significantly lower response accuracy for SCZ and BPD patients compared to HC (SCZ–HC t = 4.04, $p_{bonf} < 0.001$, d = 0.375, BPD–HC t = 3.675, $p_{bonf} = 0.002$, d = 0.341) and a Group × Cue Type interaction (F(3,108) = 10.981, p < 0.001, $\eta^2 = 0.219$) showing that HC and MDD patients showed higher response accuracy with regard to the non-social cue, whereas BPD patients showed the opposite pattern (cf. S6 Table and S2 Fig for all results).

Parameter recovery showed that all parameters could be recovered well with the exception of ω_{3card} (cf. S4 Fig).

Dynamic learning rates-second level

For the averaged precision weights (i.e., dynamic learning rates) for learning about the social $q(\psi_{2gaze})$ and non-social $q(\psi_{2card})$, we found a main effect of task phase (F(1,108) = 24.868, p < 0.001, $\eta^2 = 0.182$), showing that $q(\psi_2)$ is higher in volatile compared to the stable phases (t = -5.147, $p_{\text{bonf}} < 0.001$, d = -0.478) (Fig 4A). In the model-agnostic regression analysis, we observed increased slopes of beta weights over time delays in the volatile vs. stable phase (S5 Fig) indicating higher weighting of most recent trials, which concurs with the increased learning rates in the volatile phase in the HGF. While this effect was observable for both cues, the difference in the model-agnostic analysis was significant only for the card cue, perhaps because this analysis does not consider individual variations in cue use.



Fig 4. Results for mixed ANOVA using precision weights for updating beliefs about social and non-social contingency and volatility. (A) Precision weights $q(\psi_2)$ and (B) precision weights ψ_3 . Overall, $q(\psi_2)$ and ψ_3 increase when transitioning from stable to volatile phase. Patients with BPD show reduced overall $q(\psi_2)$. At same time, patients with BPD show higher ψ_3 compared to the other groups and a more pronounced increase in response to volatility. Bars indicate SEM. See also S1 Fig, S7 Table and S8 Table for all results.

https://doi.org/10.1371/journal.pcbi.1008162.g004

There was no significant interaction between Phase and Information Type, which indicates that $q(\psi_2)$ increases similarly during social and non-social volatility (F(1,108) = 0.131, p = 0.718, $\eta^2 = 0.001$). There was a significant main effect of group (F(3,108) = 3.557, p = 0.017, $\eta^2 = 0.088$), and the post-hoc *t*-tests revealed that participants with BPD showed significantly lower precision weights on the second level compared to HC (t = 3.101, $p_{\text{bonf}} = 0.015$, d = 0.288). The difference in $q(\psi_2)$ between groups was not affected by Information type (F(3,108) = 1.038, p = 0.379, $\eta^2 = 0.027$) or its interaction with Phase (F(3,108) = 0.940, p = 0.424, $\eta^2 = 0.025$) (see S7 Table for all results and for results with the reduced sample).

Dynamic learning rates-third level

We found a main effect of task phase on precision weights at the third level (F(1,108) =116.206, p < 0.001, $\eta^2 = 0.462$), showing that ψ_3 is higher in volatile compared to the stable phases (t = -9.784, $p_{\text{bonf}} < 0.001$, d = -0.908) (Fig 4B). There was a significant main effect of group (F(3,108) = 6.530, p < 0.001, $\eta^2 = 0.141$), and post-hoc t-tests showed that participants with BPD showed significantly higher precision weights at the third level compared to all other groups (BPD-HC *t* = -4.204, *p*_{bonf} < .001, *d* = -0.390; BPD-MDD *t* = -3.199, *p*_{bonf} = .011, *d* = -0.297; BPD–SCZ t = -3.055, p_{bonf} = .017, d = -0.284). In addition, there was a significant Phase × Group interaction (F(3,108) = 5.962, p < 0.001, $\eta^2 = 0.071$) showing that participants with BPD increase their precision weights for both modalities significantly more compared to the other groups when volatility increases. There was a trend of BPD patients showing stronger increases in ψ_3 in response to social compared to non social volatility (F(3,108) = 2.620, p = 0.055, $\eta^2 = 0.065$). The analysis also revealed that ψ_3 were affected by the order of schedule $(F(1,108) = 5.008, p = 0.027, \eta^2 = 0.036)$, with ψ_3 higher for participants receiving the incongruent-first schedule (i.e gaze starts of being highly misleading) compared to the congruentfirst schedule (i.e gaze starts of being highly helpful). This effect was not modulated by Group $(F(3,108) = 2.067, p = 0.109, \eta^2 = 0.045)$ (see S8 Table for all results and for results with the reduced sample).

Social weighting

The parameter ζ was a measure of the weight given to the social prediction relative to the learned non-social prediction (cf. Fig 5B & 5C for simulation results). Since ζ was restricted to the positive domain, estimate distributions were analyzed log-space, where they were less skewed. We found significant group differences in log(ζ) (*F*(3,108) = 5.893, *p*>0.001, η^2 = 0.130 (Fig 5A). Both patients with BPD and patients with SCZ showed significantly higher ζ estimates compared to controls (BPD: t = -3.416, $p_{\text{bonf}} = 0.005$; d = -0.847, SCZ: t = -2.855, $p_{\text{bonf}} = 0.031, d = -0.691$) but only patients with BPD differed significantly from participants with MDD (BPD: *t* = -3.003, *p*_{bonf} = 0.02, *d* = -0.818; SCZ: *t* = -2.451, *p*_{bonf} = 0.095, *d* = -0.650). Patients with MDD did not show any significant differences compared to controls (t = -0.335, $p_{\text{bonf}} = 1, d = -0.095$). There was a significant main effect of schedule (F(1,108) = 8.259, p = 0.005, $\eta^2 = 0.061$), showing that participants receiving the congruency-first schedule had higher ζ compared to participants receiving the incongruency-first schedule (t = -2.874, p_{bonf} = 0.005, d = -0.505). There was no significant interaction between Group and Schedule (F (3,108) = 0.807, p = 0.493, $\eta^2 = 0.018$). To ensure that the shared mechanism of social overweighting in BPD and SCZ was not confounded with medication or education status (cf. S1 and S2 Tables), we subjected ζ to an ANCOVA with ζ and Chlorpromazine Equivalence Units and Years of School as covariates. The effects were robust to this (cf. S9 Table for results of full and reduced sample).



Fig 5. Social weighting factor log(ζ). (A) Patients with BPD gave the social information significantly more weight compared to HC and patients with MDD. Patients with SCZ also had higher ζ compared to HC. Boxes mark 95% confidence intervals and vertical lines standard deviations. (B), Simulation results show the impact of varying weighting factor log(ζ) on combined belief b^(t) (see methods Eq 1). The combined belief b^(t) was simulated for agents with same perceptual parameters but different ζ values (highest values (log(ζ) = 5) coded in dark blue, lowest values (log(ζ) = -5) in green). (B) shows that the combined belief b^(t) of agents with high ζ values is aligned with the social input structure (blue dots) whereas these agents show a stochastic belief structure with regard to the non-social input structure (green dots) in Panel C. Conversely, agents with low ζ values show a belief structure closely aligned to the non-social input structure (C), and a stochastic belief structure with regard to the social input structure (C), and a stochastic belief structure with regard to the social input structure (C), and a stochastic belief for all results.

https://doi.org/10.1371/journal.pcbi.1008162.g005

Social Anhedonia

There was a significant difference in the interpersonal pleasure (ACIPS) ratings between the groups (F(3,103) = 5.719, p < .001) (cf. S1 Table). Post hoc *t* tests revealed that HC showed significantly higher ACIPS scores compared to patients with MDD (t = 3.088, $p_{bonf} = .016$) and with BPD (t = 3.802, $p_{bonf} = .001$) but not with SCZ (t = 1.833, $p_{bonf} = .418$). No significant differences were observed between patients with MDD and SCZ (t = -1.322, $p_{bonf} = 1$), patients with MDD and BPD (t = 0.596, $p_{bonf} = 1$), nor patients with SCZ and BPD (t = 1.978, $p_{bonf} = 0.304$). The multivariate regression using $log(\zeta)$ and social learning rate ω_{2gaze} as predictors for ACIPS scores did not show any significant results ($R^2 = 0.009$, F(2,106) = 0.477, p = 0.622).

Discussion

This study aimed to improve our understanding of the computational mechanisms that underlie the profound interpersonal difficulties in major psychiatric disorders. To achieve this, we used a probabilistic learning task in conjunction with hierarchical Bayesian modeling transdiagnostically in patients with MDD, SCZ, BPD, and healthy controls. The task required participants to perform association learning about non-social contingencies in the presence of a social cue. This allowed us to characterize and quantify the computational aspects of aberrant social inference and decision-making at an individual level. We found that patients with SCZ and BPD showed significantly poorer performance compared to HC and patients with MDD. Patients with MDD performed comparably well to HC. In addition, patients with BPD showed greater response accuracy in the social compared to the non-social domain during the stable phase, while HC and MDD patients showed the opposite pattern. This is particularly remarkable in light of their overall poorer performance. In effect, BPD patients gave up a possible reward advantage by concentrating their learning efforts disproportionately in the social domain. In addition, we found a tendency in BPD patients to follow the gaze more during volatile phases of low accuracy compared to MDD patients, which however did not reach statistical significance (Fig 3).

These findings raise the question which mechanisms underlie these patterns of behavior. In particular, they call for an investigation of the learning and decision-making mechanisms which give rise to them. Here, computational modeling enabled insights into how beliefs are updated and how these beliefs are translated into decisions: With regard to learning, we found that BPD patients showed increased precision weighting of prediction errors when learning about volatility in both non-social and social information and a tendency for even higher precision weights when learning about social compared to non-social volatility. While volatility learning rates (ψ_3) were increased in BPD compared to HC, contingency learning rates (ψ_2) were reduced compared to HC both in the social and non-social domain. This accords with a previous finding of blunted social and non-social learning in BPD [19], which was conjectured to result from aberrant volatility beliefs, causing an impairment at detecting contingency changes needed for accurate inference. Because our modeling approach was specifically designed to model beliefs about volatility, it allowed us to address this conjecture. Indeed, our data indicate that impaired contingency learning in BPD is associated with exaggerated learning about environmental volatility. A similar pattern has been observed in autism spectrum disorder (ASD) [8]. Aberrant volatility beliefs in BPD have been suggested to result from unpredictable early-life relationships [19]. However, this is a less likely explanation in ASD which is characterized as a pervasive developmental disorder. This points to a different origin of the mechanistic overlap between our findings and those of [8]. The commonality of aberrant volatility learning may explain the repeated finding of high autism quotient (AQ) values in BPD patients [8], which is confirmed in our sample (cf. S1 Table) and could be taken to suggest a partially shared mechanism of aberrant social inference in these disorders. A previous study from our group found that healthy participants scoring high on AQ showed a similar pattern to the present study's BPD patients, in that they followed the gaze more than low-AQ participants during periods of low accuracy in volatile phases [20]. In that study, computational modeling showed that high AQ participants failed to use the social information to adapt the precision of their belief about the non-social cue. However, this was not found in any participant group in the present study.

This has been the second study demonstrating that aberrant learning in BPD not only concerns social, but also non-social information (cf. [19]). This suggests that aberrant learning occurs independent of domain, in line with previous findings that precision-weighted prediction errors are computed in similar brain regions, irrespective of domain [22,41].

Unlike previous studies on reward [45–47] or volatility [15] learning in SCZ and healthy subjects at risk for psychosis [16], we did not find significant differences between SCZ and HC in this regard. The same applies to MDD patients, where one possible explanation for this negative finding is the lack of punishment for incorrect choices in our task since recent findings converge on impaired aversive learning in depression (e.g. [48,49]).

With regard to decision-making, computational modeling revealed that SCZ and BPD patients both weighted their social-domain predictions more strongly than HC and MDD. This explains the lower performance of BPD and SCZ patients. Their stronger reliance on social cues compared to HC and MDD patients was detrimental because the social cue was more volatile than the non-social one (5 as opposed to 3 contingency changes).

The commonality of over-weighting social-domain predictions in SCZ and BPD patients suggests itself as the decision-making aspect of a general interpersonal hypersensitivity in both conditions [50]. This is also reflected in excessive, albeit inaccurate, mental state attributions (hypermentalizing) [51–53] that constitute a shared feature of BPD and SCZ [6,51–58].

Hypermentalizing is also a possible explanation for the findings of [59], where similar modeling as in the present study showed that healthy participants at the high end of the paranoia spectrum used similar weighting of social information irrespective of whether incorrect advice was framed to be intentional or not, while low-paranoia participants reduced their social weighting when negative advice was cued to be intentional. Furthermore, a study of healthy participants by our group [29] found that stronger weighting of social over non-social predictions during decision-making was associated with increased activity in the putamen and anterior insula. In future studies, it will be interesting to investigate the involvement of these regions in excessive social-weighting and hypermentalizing in BPD and SCZ. Also, a direct comparison of patients with BPD and autism could help to unveil shared mechanisms of aberrant social inference.

In addition to looking for learning and decision-making differences between the different diagnostic groups that were defined by traditional ICD-10 criteria, we adopted a transdiagnostic perspective to investigate the relation between computational mechanisms of social learning and decision-making with ACIPS, a self-report measure of social anhedonia. Previous studies have adopted such a dimensional approach in the general population and found that patterns in aversive learning mapped onto distinct symptoms of depression, social anxiety, and compulsivity [48,59,60]. We, however, could not find any transdiagnostic association of social anhedonia with computational parameters of social learning and decision-making in our data. A possible explanation for this negative finding is that we used scores from a single questionnaire whereas previous studies applied factor analysis (e.g. [60]) on all items from several questionnaires. In addition, these studies investigated larger samples compared to our study (>400 vs. 116) and therefore had more statistical power to detect significant effects.

Limitations

We did not use a non-social cue (such as an arrow pointing to a card) as a control condition and therefore cannot fully rule out the possibility that the increased weighting of our social cue observed in BPD and SCZ reflects a more general rather than specifically social peculiarity in information processing. However, eye gaze is a very salient cue and in the paradigm, we aimed to accentuate the social quality of our cue by a clear period of eye contact with the participant before providing the cue.

A further limitation concerns the fact that most patients were in psychopharmacological treatment during data acquisition and had different degrees of disorder severity and chronicity. Furthermore, different patient groups were assessed in different clinical centers, and there was a gender imbalance in the SCZ and BPD groups.

Conclusion

By adopting a computational psychiatry approach [61–65] to data from an inference task with a social component, we show that BPD patients exhibit an aberrant pattern of learning rate adjustment when the environment becomes more volatile. Instead of quickly relearning changed contingencies, they show exaggerated volatility learning. While SCZ and MDD patients showed a tendency to the same pattern, they did not significantly differ from controls in this respect. We also show that BPD and SCZ patients rely more strongly than controls on social-domain beliefs relative to non-social-domain beliefs when making decisions even in a task where doing the opposite would have given them an advantage. Taken together, this shows that there are computational commonalities as well as differences between patient groups, which suggests some underlying mechanisms that may be shared across diagnoses. Since this approach allows for individually quantifying the severity of impairment at a mechanistic level, it has the potential to lead to diagnostic and prognostic advances. Furthermore, it points the way to possible targets for novel interventions which transcend traditional diagnostic boundaries.

Supporting information

S1 Table. Psychometric data of the participants. All quantities given as Mean ± SD. (DOCX)

S2 Table. Demographic data of the participants. All quantities given as Mean ± SD. (DOCX)

S3 Table. Prior configurations of perceptual and response model parameters. Means and variances of Gaussian priors are given in the space in which the parameter was estimated (native, log, or logit). (DOCX)

S4 Table. Within-subjects model comparison. (DOCX)

S5 Table. Mean posterior estimates of learning model and decision model parameters estimated from winning model.

(DOCX)

S6 Table. Statistics for mixed ANOVA with response accuracy (% High probability choices) from real and simulated data for stable and volatile phases (Factor Phase) of social and non-social cue (Factor Cue Type) for all groups (Factor Group) and schedules (Factor Schedule). The table shows the results for the real and simulated behavior. (DOCX)

S7 Table. Statistics for mixed ANOVA with averaged $q(\psi_2)$ during stable and volatile phases (Factor Phase) of social and non-social cue (Factor Cue Type) for all groups (Factor Group) and schedules (Factor Schedule). The table shows the results for the full and reduced sample.

(DOCX)

S8 Table. Statistics for mixed ANOVA with averaged ψ_3 during stable and volatile phases (Factor Phase) of social and non-social cue (Factor Cue Type) for all groups (Factor Group) and schedules (Factor Schedule). The table shows the results for the full and reduced sample.

(DOCX)

S9 Table. Statistics for ANCOVA with ζ and Chlorpromazine Equivalence Units and Years of School as covariate for full and reduced sample. (DOCX)

S1 Fig. Learning trajectories for one example participant. A, Precisions ψ_{3card} (red) and ψ_{3gaze} (blue) that modulate the weight on B, prediction errors δ_{2card} (red) and δ_{2gaze} (blue). C, Precision weights ψ_{2card} in red trajectory and $q(\psi_{2card})$ in red dotted trajectory. Precision weights modulate weight on D) prediction error δ_{1card} (red) and δ_{1gaze} (blue) signals. E, Dark red dots mark the input structure of the non-social information (blue correct = 1; green correct = 0) and the dotted red line represents the ground truth of this input structure. Light red dots mark the choices (blue card = 1; green card = 0). The red trajectory is the participant specific belief

trajectory about the blue card to be correct that was estimated on the basis of the choices. E, The same logic applies to the social input and response structure in blue. The posterior parameter estimates for this particular participant were $\omega_{2card} = -1.460$, $\omega_{2gaze} = -3.576$, $\omega_{3card} =$ -6.021, $\omega_{2gaze} = -6.074$, $\log(\zeta) = -2.0243$, $\log(\beta) = 2.207$, $\log it(\eta) = 0.211$. (TIF)

S2 Fig. Posterior Predictive Validity. Simulated Behavior from the posterior estimates of all participants revealed the same effects as real behavior. Boxes mark 95% confidence intervals and vertical lines standard deviations.

(TIF)

S3 Fig. Grouped individual data points showing precision weights for updating beliefs about social and non-social contingency and volatility. A, precision weights $q(\psi_2)$. B, precision weights ψ_3 . Overall, $q(\psi_2)$ and ψ_3 increase when transitioning from stable to volatile phase.

(TIF)

S4 Fig. Parameter recovery. We simulated behavioral responses based on the posterior estimates of all participants 10 times, resulting in 1160 simulations. For each subject, we calculated the average posteriors estimated from the simulated data (x-axis) and correlated (Pearson's correlations) them with the original posterior parameters estimates (y-axis), which is shown in plots a-f. The estimated parameters could be recovered well, however, the third level evolution rate (ω_{card}) could not be recovered well. (TIF)

S5 Fig. Regression model of participants choices for both cue types. Top: Regression model for choices in card space (1 = taking blue, 0 = taking green) with predictors of card accuracy (1 = blue correct, 0 = green correct) for the past 5 trials, and reward values (reward value if blue taken, reward value if green taken) for the whole task (left), stable (middle) and volatile (right) phase. The slope between predictors were calculated as a model-agnostic readout of the 'learning rate', indiciating the degree to which more recent information is weighted. A t-test was applied to compare the difference between these slopes during stable and volatile phases of the task. Slopes are increasing during volatile compared to stable phases. Below: Regression model for choices in gaze space (1 = taking advice, 0 = not taking advice) with predictors of gaze accuracy (1 = gaze correct, 0 = gaze incorrect) for the past 5 trials, and card reward values (reward value if advice taken) for the whole task (left), stable (middle) and volatile (right) phase.

(TIF)

Acknowledgments

Lara Henco would like to thank the Graduate School of Systemic Neurosciences (LMU). Dr. Andreea Diaconescu was supported by the Swiss National Foundation (PZ00P3_167952) and the Krembil Foundation. We would like to thank Lea Duerr, Nina von Aken, Mathilda Sommer and Benjamin Pross for helping with the recruitment and data collection.

Author Contributions

Conceptualization: Lara Henco, Andreea O. Diaconescu, Leonhard Schilbach.

Data curation: Lara Henco.

Formal analysis: Lara Henco, Andreea O. Diaconescu, Juha M. Lahnakoski, Christoph Mathys.

Funding acquisition: Leonhard Schilbach.

Investigation: Lara Henco, Marie-Luise Brandi, Sophia Hörmann, Johannes Hennings, Alkomiet Hasan, Irina Papazova, Wolfgang Strube.

Methodology: Lara Henco, Andreea O. Diaconescu, Leonhard Schilbach, Christoph Mathys.

Project administration: Lara Henco, Leonhard Schilbach.

Resources: Johannes Hennings, Alkomiet Hasan, Leonhard Schilbach, Christoph Mathys.

Software: Andreea O. Diaconescu, Juha M. Lahnakoski, Christoph Mathys.

Supervision: Leonhard Schilbach, Christoph Mathys.

Validation: Lara Henco, Andreea O. Diaconescu, Dimitris Bolis, Christoph Mathys.

Visualization: Lara Henco.

Writing - original draft: Lara Henco.

Writing – review & editing: Lara Henco, Andreea O. Diaconescu, Juha M. Lahnakoski, Marie-Luise Brandi, Sophia Hörmann, Johannes Hennings, Alkomiet Hasan, Irina Papazova, Wolfgang Strube, Dimitris Bolis, Leonhard Schilbach, Christoph Mathys.

References

- Schilbach L. Towards a second-person neuropsychiatry. Philos Trans R Soc B Biol Sci. 2016; 371 (1686). Available from: https://doi.org/10.1098/rstb.2015.0081
- Barkus E, Badcock JC. A transdiagnostic perspective on social anhedonia. Front Psychiatry. 2019; 10 (APR):1–15. Available from: https://doi.org/10.3389/fpsyt.2019.00216
- Blanchard JJ, Horan WP, Brown SA. Diagnostic differences in social anhedonia: A longitudinal study of schizophrenia and major depressive disorder. J Abnorm Psychol. 2001; 110(3):363–71. Available from: https://doi.org/10.1037//0021-843x.110.3.363 PMID: 11502079
- Kupferberg A, Bicks L, Hasler G. Social functioning in major depressive disorder. Neurosci Biobehav Rev. 2016; 69:313–32. Available from: https://doi.org/10.1016/j.neubiorev.2016.07.002 PMID: 27395342
- Fulford D, Campellone T, Gard DE. Social motivation in schizophrenia: How research on basic reward processes informs and limits our understanding Social motivation in schizophrenia: How research on basic reward processes informs and limits our understanding. Clin Psychol Rev. 2018; 63(May):12–24. Available from: https://doi.org/10.1016/j.cpr.2018.05.007
- Frith C. Schizophrenia and theory of mind. Psychol Med. 2004; 34:385–9. Available from: https://doi. org/10.1017/s0033291703001326 PMID: 15259823
- Fonagy P, Bateman AW. Mentalizing and borderline personality disorder. J Ment Heal. 2007; 16(1):83– 101. Available from: https://doi.org/10.1080/09638230601182045
- Lawson RP, Mathys C, Rees G. Adults with autism overestimate the volatility of the sensory environment. Nat Neurosci. 2017; 20(9):4–6. Available from: https://doi.org/10.1038/nn.4615
- 9. Mathys C. How could we get nosology from computation? Comput Psychiatry New Perspect Ment Illn. 2016; 20:121–38. Available from: https://mitpress.mit.edu/books/computational-psychiatry
- Mathys C, Daunizeau J, Friston KJ, Stephan KE. A Bayesian foundation for individual learning under uncertainty. Front Hum Neurosci. 2011; 5(May):1–20. Available from: https://doi.org/10.3389/fnhum. 2011.00039
- Mathys CD, Lomakina EI, Daunizeau J, Iglesias S, Brodersen KH, Friston KJ, et al. Uncertainty in perception and the Hierarchical Gaussian Filter. Front Hum Neurosci. 2014; 8(November):1–24. Available from: https://doi.org/10.3389/fnhum.2014.00001 PMID: 24474914
- Behrens TEJ, Woolrich MW, Walton ME, Rushworth MFS. Learning the value of information in an uncertain world. Nat Neurosci. 2007; 10(9):1214–21. Available from: <u>https://doi.org/10.1038/nn1954</u> PMID: <u>17676057</u>

- Behrens TEJ, Hunt LT, Woolrich MW, Rushworth MFS. Associative learning of social value. Nature. 2008; 456(7219):245–9. Available from: https://doi.org/10.1038/nature07538 PMID: 19005555
- Sterzer P, Adams RA, Fletcher P, Frith C, Lawrie SM, Muckli L, et al. The Predictive Coding Account of Psychosis. Biol Psychiatry. 2018;1–10. Available from: https://doi.org/10.1016/j.biopsych.2018.05.015
- Deserno L, Boehme R, Mathys C, Katthagen T, Kaminski J, Stephan KE, et al. Volatility Estimates Increase Choice Switching and Relate to Prefrontal Activity in Schizophrenia. Biol Psychiatry Cogn Neurosci Neuroimaging. 2020 Feb 1; 5(2):173–83. Available from: <u>https://doi.org/10.1016/j.bpsc.2019.</u> 10.007 PMID: 31937449
- Cole DM, Diaconescu AO, Pfeiffer UJ, Brodersen KH, Mathys CD, Julkowski D, et al. Atypical processing of uncertainty in individuals at risk for psychosis. NeuroImage Clin. 2020; 26(December 2019). Available from: https://doi.org/10.1016/j.nicl.2020.102239
- Reed EJ, Uddenberg S, Suthaharan P, Mathys CD, Taylor JR, Groman SM, et al. Paranoia as a deficit in non-social belief updating. Elife. 2020 May 26; 9. Available from: https://doi.org/10.7554/eLife.56345
- Petzschner FH, Weber LAE, Gard T, Stephan KE. Computational Psychosomatics and Computational Psychiatry: Toward a Joint Framework for Differential Diagnosis. Biol Psychiatry. 2017; 82(6):421–30. Available from: https://doi.org/10.1016/j.biopsych.2017.05.012 PMID: 28619481
- Fineberg SK, Leavitt J, Stahl DS, Kronemer S, Landry CD, Alexander-Bloch A, et al. Differential Valuation and Learning From Social and Nonsocial Cues in Borderline Personality Disorder. Biol Psychiatry. 2018; 84(11):838–45. Available from: https://doi.org/10.1016/j.biopsych.2018.05.020 PMID: 30041970
- Sevgi M, Diaconescu AO, Henco L, Tittgemeyer M, Schilbach L. Social Bayes: Using Bayesian Modeling to Study Autistic Trait–Related Differences in Social Cognition. Biol Psychiatry [Internet]. 2020; 87 (2):185–93. Available from: https://doi.org/10.1016/j.biopsych.2019.09.032 PMID: 31856957
- Diaconescu A, Mathys C, Weber LAE, Daunizeau J, Kasper L, Lomakina EI, et al. Inferring on the intentions of others by hierarchical bayesian learning. PLoS Comput Biol. 2014; 10(9):e1003810. Available from: https://doi.org/10.1371/journal.pcbi.1003810 PMID: 25187943
- Diaconescu A, Mathys C, Weber LAE, Kasper L, Mauer J, Stephan KE. Hierarchical prediction errors in midbrain and septum during social learning. Soc Cogn Affect Neurosci. 2017; 12(4):618–34. Available from: https://doi.org/10.1093/scan/nsw171 PMID: 28119508
- Adams RA, Brown HR, Friston KJ. Bayesian inference, predictive coding and delusions. Avant. 2015; V (3):51–88. Available from: https://doi.org/10.12849/50302014.0112.0004
- Hauser TU, Iannaccone R, Ball J, Mathys C, Brandeis D, Walitza S, et al. Role of the medial prefrontal cortex in impaired decision making in juvenile attention-deficit/hyperactivity disorder. JAMA Psychiatry. 2014; 71(10):1165–73. Available from: <u>https://doi.org/10.1001/jamapsychiatry.2014.1093</u> PMID: 25142296
- Powers AR, Mathys C, Corlett PR. Pavlovian conditioning–induced hallucinations result from overweighting of perceptual priors. Science. 2017; 357(August):596–600. Available from: https://doi.org/10. 1126/science.aan3458
- DeBerker AO De, Rutledge RB, Mathys C, Marshall L, Cross GF, Dolan RJ, et al. Computations of uncertainty mediate acute stress responses in humans. Nat Commun. 2016; 7:1–11. Available from: https://doi.org/10.1038/ncomms10996
- Bernardoni F, Geisler D, King JA, Javadi AH, Ritschel F, Murr J, et al. Altered Medial Frontal Feedback Learning Signals in Anorexia Nervosa. Biol Psychiatry. 2018 Feb 1; 83(3):235–43. Available from: https://doi.org/10.1016/j.biopsych.2017.07.024 PMID: 29025688
- Bolis D, Schilbach L. Beyond one Bayesian brain: Modeling intra-and inter-personal processes during social interaction: Commentary on "mentalizing homeostasis: The social origins of interoceptive inference" by Fotopoulou & Tsakiris. Neuropsychoanalysis. 2017; 19(1):35–8. Available from: <u>https://doi.org/10.1080/15294145.2017.1295215</u>
- Henco L, Brandi M-L, Lahnakoski JM, Diaconescu AO, Mathys C, Schilbach L. Bayesian modelling captures inter-individual differences in social belief computations in the putamen and insula. Cortex. 2020; (May):1–16. Available from: https://doi.org/10.1016/j.cortex.2020.02.024
- Baron-Cohen S, Wheelwright S, Skinner R, Martin J, Clubley E. The Autism-Spectrum Quotient (AQ): Evidence from Asperger Syndrome/High-Functioning Autism, Males and Females, Scientists and Mathematicians. J Autism Dev Disord. 2001; 31(1):5–17. Available from: https://doi.org/10.1023/ a:1005653411471 PMID: 11439754
- **31.** Gooding DC, Pflum MJ. The assessment of interpersonal pleasure: Introduction of the Anticipatory and Consummatory Interpersonal Pleasure Scale (ACIPS) and preliminary findings. Psychiatry Res. 2014; 215(1):237–43. Available from: https://doi.org/10.1016/j.psychres.2013.10.012 PMID: 24210182
- Kay SR, Fiszbein A OL. The Positive and Negative Syndrome Scale for schizophrenia. Schizophr Bull. 1987; 13(2):261–76. Available from: https://doi.org/10.1093/schbul/13.2.261 PMID: 3616518

- Benkert O, Müller MJ, Schlösser R, Addington D, Wetzel H, Marx-Dannigkeit P. The Calgary Depression Rating Scale for Schizophrenia: development and interrater reliability of a German version (CDSS-G). J Psychiatr Res. 2002; 33(5):433–43. Available from: https://doi.org/10.1016/s0022-3956(99) 00018-7
- Bohus M, Kleindienst N, Limberger MF, Stieglitz RD, Domsalla M, Chapman AL, et al. The short version of the Borderline Symptom List (BSL-23): Development and initial data on psychometric properties. Psychopathology. 2009; 42(1):32–9. Available from: https://doi.org/10.1159/000173701 PMID: 19023232
- Stoet G. PsyToolkit: A Novel Web-Based Method for Running Online Questionnaires and Reaction-Time Experiments. Teach Psychol. 2017; 44(1):24–31. Available from: https://doi.org/10.1177/0098628316677643
- Daunizeau J, den Ouden HEM, Pessiglione M, Kiebel SJ, Stephan KE, Friston KJ. Observing the observer (I): Meta-bayesian models of learning and decision-making. PLoS One. 2010; 5(12). Available from: https://doi.org/10.1371/journal.pone.0015554
- Daunizeau J, den Ouden HEM, Pessiglione M, Kiebel SJ, Friston KJ, Stephan KE. Observing the Observer (II): Deciding When to Decide. PLoS One. 2010; 5(12). Available from: <u>https://doi.org/10.1371/journal.pone.0015555</u>
- Jara-Ettinger J. Theory of mind as inverse reinforcement learning. Curr Opin Behav Sci. 2019; 29:105– 10. Available from: https://doi.org/10.1016/j.cobeha.2019.04.010
- Sutton RS. Gain Adaptation Beats Least Squares? Proc Seventh Yale Work Adapt Learn Syst. 1992;161–166. Available from: papers://d471b97a-e92c-44c2-8562-4efc271c8c1b/Paper/p596
- Rescorla RA, Wagner AR. A Theory of Pavlovian Conditioning: Variations in the Effectiveness of Reinforcement and Nonreinforcement. 1972;1–18. Available from: papers2://publication/uuid/51EED98C-39D3-4ECA-9CC8-F7E445CCB145
- Iglesias S, Mathys C, Brodersen KH, Kasper L, Piccirelli M, denOuden HEM, et al. Hierarchical Prediction Errors in Midbrain and Basal Forebrain during Sensory Learning. Neuron. 2013; 80(2):519–30. Available from: https://doi.org/10.1016/j.neuron.2013.09.009 PMID: 24139048
- Stephan KE, Penny WD, Daunizeau J, Moran RJ, Friston KJ. Bayesian model selection for group studies. Neuroimage. 2009; 46(4):1004–17. Available from: https://doi.org/10.1016/j.neuroimage.2009.03. 025 PMID: 19306932
- Rigoux L, Stephan KE, Friston KJ, Daunizeau J. Bayesian model selection for group studies—revisited. Neuroimage. 2014; 84:971–85. Available from: <u>https://doi.org/10.1016/j.neuroimage.2013.08.065</u> PMID: 24018303
- Wilson RC, Collins AGE. Ten simple rules for the computational modeling of behavioral data. Elife. 2019; 8:1–33. Available from: https://doi.org/10.7554/eLife.49547
- 45. Gradin VB, Kumar P, Waiter G, Ahearn T, Stickle C, Milders M, et al. Expected value and prediction error abnormalities in depression and schizophrenia. Brain. 2011; 134(6):1751–64.
- 46. Juckel G, Schlagenhauf F, Koslowski M, Wüstenberg T, Villringer A, Knutson B, et al. Dysfunction of ventral striatal reward prediction in schizophrenia. Neuroimage. 2006; 29(2):409–16. Available from: https://doi.org/10.1016/j.neuroimage.2005.07.051 PMID: 16139525
- Waltz JA, Frank MJ, Robinson BM, Gold JM. Selective reinforcement learning deficits in schizophrenia support predictions from computational models of striatal-cortical dysfunction. 2008; 62(7):756–64. Available from: https://doi.org/10.1016/j.biopsych.2006.09.042
- 48. Wise T, Dolan RJ. Associations between aversive learning processes and transdiagnostic psychiatric symptoms revealed by large-scale phenotyping. bioRxiv [Internet]. 2019; 44(0):843045. Available from: https://www.biorxiv.org/content/10.1101/843045v1
- 49. Mukherjee D, Lee S, Kazinka R, D Satterthwaite T, Kable JW. Multiple Facets of Value-Based Decision Making in Major Depressive Disorder. Sci Rep. 2020; 10(1):3415. Available from: https://doi.org/10. 1038/s41598-020-60230-z PMID: 32099062
- Gunderson JG, Lyon-Ruth K. BPD's interpersonal hypersensitivity phenotype: a gene-environmentdevelopmental model. J Pers. 2008; 22(1):1–17. Available from: https://doi.org/10.1521/pedi.2008.22. 1.22.BPD
- Sharp C, Ha C, Carbone C, Kim S, Perry K, Williams L, et al. Hypermentalizing in adolescent inpatients: Treatment effects and association with borderline traits. J Pers Disord. 2013; 27(1):3–18. Available from: https://doi.org/10.1521/pedi.2013.27.1.3 PMID: 23342954
- 52. Fonagy P, Luyten P. A developmental, mentalization-based approach to the understanding and treatment of borderline personality disorder. Dev Psychopathol. 2009 Jan; 21(4):1355–81. Available from: https://doi.org/10.1017/S0954579409990198 PMID: 19825272

- Sharp C, Pane H, Ha C, Venta A, Patel AB, Sturek J, et al. Theory of mind and emotion regulation difficulties in adolescents with borderline traits. J Am Acad Child Adolesc Psychiatry. 2011; 50(6):563–573. e1. Available from: https://doi.org/10.1016/j.jaac.2011.01.017 PMID: 21621140
- Okruszek Ł, Haman M, Kalinowski K, Talarowska M, Becchio C, Haman M, et al. Impaired Recognition of Communicative Interactions from Biological Motion in Schizophrenia. PLoS One. 2015; 10(2): e0116793. Available from: https://doi.org/10.1371/journal.pone.0116793 PMID: 25664584
- Lynch TR, Rosenthal MZ, Kosson DS, Cheavens JS, Lejuez CW, Blair RJR. Heightened sensitivity to facial expressions of emotion in borderline personality disorder. Emotion. 2006; 6(4):647–55. Available from: https://doi.org/10.1037/1528-3542.6.4.647 PMID: 17144755
- 56. Lowyck B, Luyten P, Vanwalleghem D, Vermote R, Mayes LC, Crowley MJ. What's in a face? Mentalizing in borderline personality disorder based on dynamically changing facial expressions. Personal Disord Theory, Res Treat. 2015; 7(1):72–9. Available from: https://doi.org/10.1037/per0000144
- Corcoran R, Cahill C, Frith CD. The appreciation of visual jokes in people with schizophrenia: A study of "mentalizing" ability. Schizophr Res. 1997; 24(3):319–27. Available from: <u>https://doi.org/10.1016/</u> s0920-9964(96)00117-x PMID: 9134592
- Abu-Akel A, Bailey AL. The possibility of different forms of theory of mind impairment in psychiatric and developmental disorders. Psychological medicine. 2000; 30(3):735–38. Available from: https://doi.org/10.1017/s0033291799002123 PMID: 10883728
- Seow TXF, Gillan CM. Transdiagnostic Phenotyping Reveals a Host of Metacognitive Deficits Implicated in Compulsivity. Sci Rep. 2020; 10(1):1–11. Available from: https://doi.org/10.1038/s41598-019-56847-4 PMID: 31913322
- Gillan CM, Kosinski M, Whelan R, Phelps EA, Daw ND. Characterizing a psychiatric symptom dimension related to deficits in goaldirected control. Elife. 2016; 5(MARCH2016):1–24. Available from: https://doi.org/10.7554/eLife.11305
- Read Montague P, Dolan RJ, Friston KJ, Dayan P. Computational psychiatry. 2013; 16(1):72–80. Available from: https://doi.org/10.1016/j.tics.2011.11.018
- Stephan KE, Mathys C. Computational approaches to psychiatry. Curr Opin Neurobiol. 2014; 25:85– 92. Available from: https://doi.org/10.1016/j.conb.2013.12.007 PMID: 24709605
- Wang X-J, Krystal JH. Computational Psychiatry. Neuron. 2014; 84(3):638–54. Available from: https://doi.org/10.1016/j.neuron.2014.10.018 PMID: 25442941
- Adams RA, Huys QJM, Roiser JP. Computational Psychiatry: Towards a mathematically informed understanding of mental illness. J Neurol Neurosurg Psychiatry. 2016; 87(1):53–63. Available from: https://doi.org/10.1136/innp-2015-310737 PMID: 26157034
- Huys QJM, Maia T V, Frank MJ. Computational psychiatry as a bridge from neuroscience to clinical applications. Nat Neurosci. 2016; 19(3):404–13. Available from: https://doi.org/10.1038/nn.4238 PMID: 26906507